# Singapore Management University

# Institutional Knowledge at Singapore Management University

Research Collection School of Social Sciences

School of Social Sciences

2-2020

# Solving mate shortages: Lowering standards, searching farther, and abstaining

Peter K. JONASON Western Sydney University

Simone L. BETES Western Sydney University

Norman P. LI Singapore Management University, normanli@smu.edu.sg

Follow this and additional works at: https://ink.library.smu.edu.sg/soss\_research

Part of the Gender and Sexuality Commons, Personality and Social Contexts Commons, and the Social Psychology Commons

#### Citation

JONASON, Peter K., BETES, Simone L., & LI, Norman P.. (2020). Solving mate shortages: Lowering standards, searching farther, and abstaining. *Evolutionary Behavioral Sciences*, *14*(2), 160-172. **Available at:** https://ink.library.smu.edu.sg/soss\_research/3169

This Journal Article is brought to you for free and open access by the School of Social Sciences at Institutional Knowledge at Singapore Management University. It has been accepted for inclusion in Research Collection School of Social Sciences by an authorized administrator of Institutional Knowledge at Singapore Management University. For more information, please email <a href="mailto:cherylds@smu.edu.sg">cherylds@smu.edu.sg</a>.

# Solving mate shortages:

Lowering standards, searching farther, and abstaining

Peter K. Jonason

Western Sydney University

Norman P. Li

Singapore Management University

Simone Betes

Western Sydney University

Published in Evolutionary Behavioral Sciences. 2019. Advance online publication. https://doi.org/10.1037/ebs0000174

Creative Commons Attribution-Noncommercial-No Derivative Works 4.0 License Accepted version, April 2019

Running head: SOLVING MATE SHORTAGES

2

Abstract

Although much work on mating psychology has focused on mate preferences and responses

to desirable sexual and romantic offers, less is known about what happens when individuals

face a lack of mating options. We present two studies on (hypothetical) compensatory mating

tactics. In Study 1 (N = 299), participants were asked to imagine they were struggling to find

long-term and short-term mates and we revealed sex differences and context-specific effects

consistent with parental investment theory. In Study 2 (N = 282), participants were asked to

imagine they had been incapable of finding a short-term and long-term mate for six months

despite actively trying to find one and then report the likelihood of abstaining, lowering their

standards, and traveling farther to find a satisfactory partner; results largely (and

conceptually) replicated those from Study 1 but document the role of attachment and (self-

reported) mate value in accounting for individual differences in adopting the three mating

tactics. We frame our results in terms of how people might solve mate shortages.

**Keywords**: Sex Differences; Personality; Mating Strategies; Evolutionary Psychology

**Public Significance Statement** 

A major source of frustration in people's lives is their inability to find the romantic and sexual partners they want. This study provides an examination of three compensatory mating tactics people may use when struggling to satisfy their mating needs. Understanding the costs and benefits of these options may help guide people to make more informed decisions of what to

do when their romantic and sexual lives are not to their liking.

There is no shortage of research on mate preferences, showing that sex differences in mate preferences have not changed all that much in the last 30 years (Bech-Sørensen & Pollet, 2016), can be replicated in places like China (Chang, Wang, Shackelford, & Buss, 2011), and can be found in dating ad studies decades ago (Harrison & Saeed, 1977). Much of the debate in this area has focused on issues of mate preferences, potential sex differences, and the origin/nature of those preferences with some researchers treating mate preferences as a function of sociological or structural forces (Eagly, 2000; Eagly & Wood, 1999; Zetner & Eagly, 2015) with others contending mate preferences are vestiges of ancient selection pressures (Howard, Blumstein, & Schwartz, 1987; Li & Metzler, 2015). Whatever the origin of mate preferences, countless people are unable to maximize their mating ideals despite the apparent supply of attractive and available partners (Apostolou, 2017, 2019), suggesting people's mating psychologies are sensitive to contextual threats (Reeve, Kelly, & Welling, 2016). Indeed, the whole online dating industry is built on people's desire and struggle to find mates. Tinder<sup>®</sup> (the leading mobile dating application; Duguay, 2016) alone has over 10 million downloads daily, suggesting an examination of what people do when they cannot find mates is warranted. Mate shortages can come about from being characterized by too many "dealbreakers" (Jonason, Garcia, Webster, Li, & Fisher, 2015), the quality of alternatives (Gutierres, Kenrick, & Partch, 1999), or living in a biased operational sex ratio (Moss & Maner, 2015; Schacht & Borgerhoff Mulder, 2015). When faced by shortages, people should have adaptive responses to mate shortages, or compensatory mating tactics. In this study, we examine three compensatory mating tactics, sex differences in their use, and the role of mate value and attachment in their use.

Previous work on compensatory mating tactics suggests that men's attitudinal and behavioral promiscuity (i.e., sociosexuality) is sensitive to changes in the availability of mates (Arnocky, Woodruff, & Schmitt, 2016), the perception of the availability of mates

encourages jealousy, mate guarding, and intrasexual competition in both sexes (Arnocky, Ribout, Mirza, & Knack, 2014), and mate preferences in both sexes are sensitive to partner scarcity (Taylor, 2013; Watkins, Jones, Little, Debruine, & Feinberg, 2012). Understanding how people solve mate shortages may inform researchers as to individual differences in mating psychology. It may also inform the public of potential strategies they may choose from when struggling to find a mate along with their related costs. We expect people's compensatory mating tactics to be sensitive to costs and benefits of choosing different compensatory mating tactics as a function of participant's sex, the level of investment in the relationship, their mate value, and attachment systems

From an evolutionary perspective, sex differences in mating strategies is a function of the size of the sex cells of each sex which leads to physiological and behavioral differences in the minimum obligatory investment each sex must make in their offspring (Trivers, 1972). Because mammalian males have abundant and small sex cells (i.e., sperm) they invest much less into a mating than females, who have scarce and large sex cells (i.e., ova). This parental investment theory has been modified in human mating psychology to point out that investment in relationships moderates sex differences such that when both sexes invest lots (i.e., long-term relationships) men and women's mating psychology should be similar whereas when only women invest lots (i.e., short-term relationships), the sexes will have different mating psychologies (Buss & Schmitt, 1993). For example, both sexes want a kind mate for long-term partnerships (Li, Bailey, Kenrick, & Linsenmeier, 2002), but men place a higher premium on physical attractiveness in short-term relationships than women do (Li & Kenrick, 2006). Such context-specific sex differences should manifest themselves in compensatory mating tactics and further be sensitive to dispositional biases in mate value and attachment to better understand why men and women might choose one compensatory mating tactic or another.

# The Current Study

We consider three potential solutions to mating shortages; lowering standards (Regan, 1998), abstention (Apostolou, 2017, 2019), and searching farther (Jonason, Nolland, & Tyler, 2017). Each of these comes with different costs and benefits that may appeal to men and women for reasons consistent with evolutionary reasoning. Lowering one's standards or "settling" on lower quality mates may increase the total pool of candidate mates available to a person by lowering minimum thresholds for acceptability but may result in pairings with suboptimal or even problematic mates (Jonason et al., 2015). People may settle on lower quality mates out of fear of being single (Spielmann et al., 2013). Abstention, in contrast, may come with losses for mating opportunities now but can result in potentially superior future matings (Apostolou, 2019. And last, traveling farther to find mates comes with metabolic travel costs, time lost in searching, and risk involved with trying to securing and engaging in outgroup matings (e.g., violence, sexual infections) but will increase the pool of candidate mates by searching a wider physical range. Prior research has examined traveling as a compensatory mating tactic in female fur seals (Callorhinus ursinus; Hoffman, Forcada, Trathan, & Amos, 2007) and it might be a viable, adaptive (yet understudied in people) tactic to find mates in humans, if we can infer what people might do based on understanding what non-humans have done to solve similar problems (Wilkins & Ebach, 2014).

Sex Differences and Similarities Across Contexts. The way people balance the associated costs and benefits with the choice of tactics may be revealed through an examination of sex differences across the long-term and short-term contexts (Buss & Schmitt, 1993). Because women may have evolved to be less willing to settle for lower quality mates than men are (Feingold, 1992; Kenrick, Groth, Trost, & Sadalla, 1993; Li, Valentine, & Patel, 2011), we expect women to be unwilling to lower their standards for both long- and short-term mates (H1a), but because ancestral men may have benefitted from a more opportunistic

mating pattern (Symons, 1979; Trivers, 1972), men should be more willing to lower their short-term standards (more than women and more than men do for long-term mates) as a mating tactic (H1b). In contrast, we predict that women should be more likely to abstain than men are (H2a) because it helps them avoid mating costs (Jonason et al., 2015) and that abstention (versus lowering standards) should be a particularly appealing tactic in long-term contexts (H2b) because of the heavy investment both sexes give in this context. And last, we expect searching farther<sup>1</sup> as a tactic to increase the number of mates one is exposed to be particularly appealing in the long-term mating context for men and women (H3a) because the costs/time will be unappealing when one wants immediate sexual gratification (i.e., the primary function people ascribe to casual sex; Jonason, 2013) and it may allow people to maintain selectivity while increasing the probability they will find someone, but women may be less willing to travel for their short-term mates given (H3b) because they are less proactive in mate searching than men are.

The Role of Mate Value and Attachment. Beyond sex differences and context effects, we explore the role of two key individual differences; mate value and attachment patterns. Mate value (Buss & Shackelford, 2008; Fisher, Cox, Bennett, & Gavric, 2008; Kirsner, Figueredo, & Jacobs, 2003) reflects one's relative bargaining power in the mating market to maximize their mating goals (Fletcher, Tither, O'Loughlin, Friesen, & Overall, 2004) and, therefore, may guide an individual's choice of and ability to pursue particular mating strategies. In addition, attachment patterns may play a role in choosing compensatory mating tactics. Attachment is a set of relationship expectancies derived from childhood experiences (Ainsworth, Blehar, Waters, & Wall, 1978; Hazan & Shaver, 1987), may play a

<sup>&</sup>lt;sup>1</sup> To date, the role of distance in mate selection has received limited attention – most of the work has been about how individuals cope with the distance in ongoing relationships (Carpenter & Knox, 1996; Feeney, 1999; Sahlstein, 2004), in another name—propinquity—to predict who marries whom (Bossard, 1932), and in nonhumans including fur seals (*Callorhinus ursinus*; Hoffman et al., 2007), colonial lesser kestrels (*Falco naumanni*; Calabuig, Ortego, Cordero, & Aparicio, 2008), and chacma baboons (*Papio ursinus*; Clarke et al., 2008).

role in decision-making in relationships (Milkulincer & Shaver, 2003; Simpson, 1990) beyond broader units of personality (Shaver & Brennan, 1992) and account for variance in engaging in casual sex around the world (Brennan & Shaver, 1995; Schmitt & Jonason, 2015). In relation to lowering standards, we expect the people who are anxiously attached (H1c) and low in mate value (H1d) to be willing to lower their standards (regardless of context) as these individual differences may reflect individual differences motivating people to not want to wait and risk being lonely (Schachner & Shaver, 2004; Simpson, 1990); they may fear being single (Spielmann et al., 2013). In relation to abstention, we expect those who felt they have more mate value may be more willing to abstain and wait for better mating options to come along as opposed to other options (H2c), because they believe as a function of their (self-perceived) mate value they "can do better" or can afford to wait (Regan, 1998). And when considering casting a wider net, we expect that people low (versus high) in mate value should be more willing to travel for their relationships (regardless of context), because this sense of low worth may motivate people to do more, including traveling farther, to find the love they desire (H3c). Indeed, in male chacma baboons (Papio ursinus; Clarke et al., 2008), the distance a male must search for sexual partners is a function of his mate value (i.e., rank).

Perhaps there is no more fundamental problem for the lovelorn and lonely than finding desirable long-term and short-term mates. Failure to do so will adversely affect the relationship/sexual satisfaction and evolutionary fitness of the individual. As such, people may adopt a variety of solutions to this problem; solutions that come with different costs and benefits. While prior research has examined the role of lowering standards (Regan, 1998) and abstaining (Apostolou, 2017) in response to failures to finding mates, little work has examined the alternative of searching farther and no work has examined these three options simultaneously. In two studies, we examine the appeal of these three compensatory tactics,

examine sex differences and context effects, and explore the role of individual differences in attachment and (self-reported) mate value in understanding who adopts different strategies.

#### Study 1: Context effects and sex differences in tactical choices

We started our investigation of people's mating tactics with a survey employing a forced-choice method. By forcing people to choose between options, this method reveals their decision-making processes differently than Likert-style methods (Li et al., 2002; Li & Kenrick, 2006). We juxtaposed the choice in three potential compensatory mating tactics and examined the role of participant's sex and relationship duration in accounting for choices.

#### Method

# Participants and procedure

Participants were 299 (45% female) paid Mechanical Turk (US\$0.50) workers aged 18-61 (M=32.56, SD=9.44) from the USA.<sup>2</sup> The minimum sample size was set at 250 because correlations stabilize at that sample size (Schönbrodt & Perugini, 2013). The majority (56%) of participants were in some form of serious relationship (including marriage)<sup>3</sup>, were heterosexual (94%), and had experience in long distance relationships in the past (64%). Only participants from unique IP addresses were included. Ethics approval was granted by XXXXX University. Ethnically, participants self-identified as white/European (72%), Asian (12%), Hispanic/Latino (9%), African/Black (5%), and "other" (2%). These participants were informed of the nature of the study. If they gave consent, they advanced through a series of randomized self-report personality questions (intended for elsewhere),

<sup>&</sup>lt;sup>2</sup> There were no MTruk controls but, we only sampled "Master" workers.

<sup>&</sup>lt;sup>3</sup> There were no differences between single people compared to those who were involved in a relationship in what they would do in a short-term relationship ( $\chi^2 = 3.46$ ) but there were differences in choices across relationship status ( $\chi^2 = 7.92$ , p < .05) with no difference in willingness to lower standards, committed people (n = 157) being more willing to choose traveling compared to single people (n = 124), and single people (n = 68) choosing abstention more than committed people (n = 46). We treat this as an anomalous effect and do not consider relationship status further.

reported mating tactics responses to mating shortages, and provided demographic details.

Upon completion, they were thanked and debriefed.

#### Measure

To measure mating tactic choices, we asked participants to "imagine you are struggling to find a prospective long-term [short-term] partner, please select the option below that is most appealing to you." In a randomized fashion, participants were presented with an option for traveling farther (i.e., "I would rather increase the distance I am willing to travel."), lowering standards (i.e., "I would rather lower my standards."), and for abstention (i.e., "I would abstain from such a relationship at this time."). Participants were provided with definitions of *short-term partners* as "a casual sex partner, someone you might have a one-night stand with or a booty-call relationship with" and *long-term partners* as "a serious relationship partner, someone you would call boyfriend/girlfriend and even consider marrying" for standardization purposes. Participants were asked to choose just one of these three options and if they were in a relationship to answer as if they were single.

#### **Results**

Descriptive statistics are reported in Table 1. In the long-term context, people preferentially chose the travel option followed by abstention, and were least willing to lower their standards ( $\chi^2[2] = 32.49$ , p < .01). In contrast, in the short-term context, abstaining was the preferred option, followed by traveling farther, and lowering standards was again the least appealing option ( $\chi^2[2] = 10.06$ , p < .01). When we looked at whether men and women chose among these options differently, we found significant effects for women in both the long-term ( $\chi^2[2] = 57.15$ , p < .01) and short-term mate ( $\chi^2[2] = 58.40$ , p < .01) shortages but only for long-term shortages in men ( $\chi^2[2] = 104.95$ , p < .01). This suggests that it is in the short-term context that the sexes may differ and that men may be somewhat ambivalent to strategy choices when then they face a short-term mate shortage, perhaps as a function of their

opportunistic mating strategy. For the long-term shortage, men preferred to travel farther (Residual = 86) and not to lower their standards (Residual = -58) or abstain (Residual = -28). Women, when facing a short-term mating shortage, preferred abstaining (Residual = 39.3), wanted to avoid lowering their standards (Residual = -31.7), and were relatively ambivalent to traveling farther (Residual = -7.7). Women, when facing a long-term mating shortage, preferred to travel farther (Residual = 40.3) and wanted to avoid lowering their standards (Residual = -27.7) and (somewhat) abstention (Residual = -12.7).

We further tested whether there were context-dependent sex differences in these choices (see Table 1). There were no sex differences in the adoption of different compensatory mating tactics in the long-term context. There was, however, a sex difference in the short-term mating context. While the sexes did not differ in their willingness to search farther for a short-term relationship, when women faced a hypothetical short-term mate shortage, they were likely to choose abstention (Residual = 4.5) and especially unlikely to lower their standards (Residual = -3.8) whereas when men face a hypothetical short-term mate shortage, they were likely lower their standards (Residual = 2.4) and unlikely to abstain (Residual = -2.9).

#### **Discussion**

This study represents the first examination of three compensatory mating tactics. Men and women appear to prefer to travel farther for long-term relationships. This may be because they are unwilling to tolerate the costs for (especially) lowering their standards and (somewhat) abstaining. When it comes to short-term relationships, women were more likely to abstain whereas men were more likely to lower their standards. Results align with parental investment (Trivers, 1972) and sexual strategies (Buss & Schmitt, 1993) theory predictions related to cost asymmetries in mating choices in men and women as a function of the duration of the courtship.

# **Study 2: Responding to Shortages**

In Study 1 we found that the selection of mating tactics varied according to sex and mating duration as expected. However, we relied solely on forced-choice questions which might artificially inflate sex differences. In Study 2, we replicate Study 1 by explicitly asking participants to imagine they are experiencing short-term and long-term mating droughts and to indicate how likely they are to adopt three mating tactics. In addition, we extend this by exploring the role of two salient individual differences; mate value and attachment.

# Method

# Participants and procedure

Participants were 282 (59% female, 2 "other" excluded from sex-related analyses) paid Mechanical Turk (US\$0.50) workers aged 18-72 (M = 34.75, SD = 10.85) from the USA who were sampled independently and one month after Study 1.<sup>4</sup> Sample size minimums and ethical permissions were established like Study 1. The majority (74%) of participants were in some form of serious relationship (including marriage)<sup>5</sup>, were heterosexual (94%), and had experience in long distance relationships in the past (67%). Ethnically, participants self-identified as white/European (74%), Asian (9%), Hispanic/Latino (8%), African/Black (7%), and "other" (2%). Participants were informed of the nature of the study. If they gave consent, they advanced through a series of randomized (across measures) self-report personality questions, reported compensatory responses to mate shortages, and provided demographic details. Upon completion, they were thanked and debriefed.

**Measures**. We used Li's Mate Value scale (see Jonason et al., 2019) to measure three dimensions of participant's self-reported mate value. Participants were asked their agreement (1 = *Strongly disagree*; 7 = *Strongly agree*) with items presented in a randomized fashion.

<sup>&</sup>lt;sup>4</sup> There were no MTruk controls used in this study except for sampling "Master" workers.

<sup>&</sup>lt;sup>5</sup> There were only weak differences for relationship status in the adoption of the any of the mate shortage solutions, thus we proceeded with tests without taking this into consideration further.

Items were averaged to create indexes of attractiveness as a short-term mate (e.g., "Compared to my peers, I am very attractive or desirable to the opposite sex"), attractiveness as a long-term mate (e.g., "People seem to be interested in having a long-term relationship with me"), and a general unattractiveness or difficulty in relationships (e.g., "I tend to have a more difficult time attracting potential mates than other people do"). Items were averaged to create indexes of attractiveness as a short-term mate (Cronbach's  $\alpha$  = .87), attractiveness as a long-term mate ( $\alpha$  = .77), and a general undesirability or difficulty in relationships ( $\alpha$  = .90). Short-term and long-term mating attractiveness were correlated (r(280) = .37, p < .01). Long-term mating attractiveness was correlated with general undesirability (r(280) = .39, p < .01) and anxious attachment (r(280) = .31, p < .01). And general undesirability was correlated with anxious attachment (r(280) = .77, p < .01).

To measure individual differences in attachment patterns we used the Experiences in Close Relationships-Revised (ECR-R) Questionnaire (Fraley, Waller, & Brennan, 2000). Participants were asked how much they agreed ( $1 = Disagree\ Strongly$ ;  $7 = Agree\ Strongly$ ) with 36 randomized items described. For instance, to measure anxious attachment, participants were shown the item "I'm afraid that I will lose my partner's love" whereas to measure avoidant attachment, participants were shown the item "I prefer not to show a partner how I feel deep down". Items were averaged to create indexes of anxious attachment ( $\alpha = .93$ ) and avoidant attachment ( $\alpha = .82$ ) that were orthogonal (r = .08).

To assess individual differences in how participants responded to mating scarcity, they were asked to imagine an inability to find mates with the following text:

For the next questions, imagine you are having trouble finding a sexual partner for a short-term relationship [serious relationship partner for a long-term

<sup>&</sup>lt;sup>6</sup> As this is an unpublished scale, we encourage the interested reader to contact the third author to assess the face validity of the items.

relationship], you have been actively trying for 6 months. Think about how that would make you feel and answer the questions below with that in mind.

Participants were then provided with three options (i.e., Search farther in terms of distance from you; Lower your standards of how particular you are about whom you date; Make no changes at all and remain single). They were asked how likely they would be to do the following ( $1 = Not \ at \ all$ ;  $5 = Very \ Much$ ) assuming they were now single. Responses across mating context and the same tactics were correlated ( $rs \approx .60$ , ps < .01). The order of the responses and the mating durations were randomized.

# **Results**

We begin by testing two mixed-model ANOVAs with the three solutions to mating shortages within each mating duration (descriptive statistics provided in Table 2). When success in short-term mating endeavors was (imagined to be) threatened, we found a main effect across the three possible solutions (F[2, 554] = 10.60, p < .01,  $\eta_p^2 = .04$ ) such that people were most willing to abstain, least willing to lower standards, and moderately willing to travel with all comparisons significant (p < .05). We also found an interaction of participant's sex and the long-term mating threat solutions ( $F[2, 554] = 8.00, p < .05, \eta_p^2 =$ .03). When success in long-term mating endeavors was threatened, we found a main effect across the three possible solutions (F[2, 552] = 44.26, p < .01,  $\eta_p^2 = .14$ ) such that people were most willing to abstain than lower their standards (p < .01) and travel more than lower their standards (p < .01) with no difference between traveling farther and abstaining. We also found an interaction of participant's sex and the long-term mating threat solutions (F[2, 552]= 3.11, p < .05,  $\eta_p^2 = .01$ ). Both interactions above reflect differences in men and women's willingness to adopt a given strategy in response to mating threats in each domain. Women were more willing to travel farther for a long-term than a short-term partner (t = -3.71, p <.01) but more willing to lower their standards for a short-term than a long-term partner (t =

2.70, p < .01). While men and women similarly lowered their standards more for short-term partners than long-term ones (t = 4.57, p < .01), men were more likely to abstain when facing a long-term mating shortage (t = -1.99, p < .01). In addition, men were more willing to lower their standards for their short-term (t = -3.91, p < .01) and less so for their long-term partners (t = -2.20, p < .05) whereas women were more willing to abstain when faced with a short-term mate shortage (t = 1.98, p < .05).

Next, we examined individual differences in attachment and mate value (Table 3). Those who felt they had difficulty in relationships and attachment anxiety were more willing to travel and lower their standards when faced with a short-term mate shortage. Those who felt they were undesirable as mates and reported attachment anxiety indicated greater propensity to travel and lower their standards when faced with a long-term mate shortage. Attachment avoidance was associated with abstaining when faced with a short-term mate shortage. Those who felt they had high long-term mate value were unlikely to abstain when facing a long-term mate shortage, were unlikely to lower their standards when faced with a short-term mate shortage and were more likely to travel when faced with a long-term mate shortage.

When we tested whether the overall correlations were moderated by participant's sex, we found little evidence for systematic moderation. Men who were low on avoidant attachment (r = -.14, p < .05) were more willing (Fisher's z = 2.04, p < .05) to lower their standards when facing a short-term mating shortage compared to women (r = .10, ns) and the same men (r = -.14, p < .05) were more willing (z = 1.88, p < .05) to lower their standards when facing a long-term mating shortage compared to women (r = .09, ns). In contrast, women with short-term mating attractiveness (r = .14, ns) were slightly more likely to lower their standards when faced with a long-term mate shortage (z = 1.72, p < .05) compared to men (r = -.07, ns).

#### **Discussion**

Again, we have shown evidence consistent with sexual strategies theory that sex differences in mating strategies are sensitive to relationship context. There are systematic patterns in the choices men and women make in how they opt to solve the adaptive problem of mating scarcity. Women remain biased against lowering their standards in general but were especially unwilling to do so in the short-term. Men in contrast, were especially willing to lower their standards for short-term relationships. Traveling farther is an appealing option for long-term partners as it likely offsets the related costs of the other two but it is unappealing in the short-term given delays in mating, costs, and even a conflict with the sexual gratification motives behind casual sex. In addition, willingness to adopt the compensatory tactics was correlated with individual differences in mate value and attachment and further moderated (slightly) by participant's sex.

# **General Discussion**

Walk down the street of Sydney, Singapore, or Seattle and one will be presented with an apparently infinite range of potential mates. Similarly, swiping through dating applications like Tinder® presents a similar picture; there is a plethora of apparently available partners (Apostolou, 2019). However, despite the abundance, many people struggle to find romantic and sexual partners and we contend this is likely to be a recurrent adaptive problem whether in high population areas or not. Some people even fear they have run out time to find a mate (Spielmann et al., 2013). Finding mates is the first problem one must solve before being able to select a good partner, rearing offspring, and retaining the mate – tasks that are considered fundamental from an evolutionary perspective on mating psychology. In hopes of understanding individual differences in tactics adopted to solve this problem, we asked participants to choose between three mating tactics when imaging a mate shortage. We lowering standards (Regan, 1998), abstaining (Apostolou, 2017), and searching farther

(Jonason, Nolland, & Tyler, 2017), adding to prior work examining how mating strategies are sensitive to partner scarcity (Arncocky, Woodruff, & Schmitt, 2016; Arnocky et al., 2014; Taylor, 2013; Watkins et al., 2012). To understand these compensatory mating tactics, we examined sex differences in the long-term and short-term mating contexts (Buss & Schmitt, 1993) and examined the role of individual differences in (self-reported) mate value and attachment in predicting variance in opting into these tactics when faced with a mating and relationship drought.

Our results are consistent with evolutionary reasoning that suggests men and women are sensitive to the costs and benefits they each face and receive, respectively, from potential mates (Symons, 1979; Trivers, 1972). We examined the option of lowering standards (Regan, 1998), which may allow for finding more mates, but comes with the cost of pairing with less desirable partners. This is consistent with work suggesting that those who fear being single, they consistently pick partners who are less responsive and less attractive as dating targets (Spielmann et al., 2013). Men, in the short-term context, were particularly willing to choose this option as were those who have psychological indicators of a need for love in the form of attachment dysfunctions and a sense of limited success in romantic endeavors. By lowering their standards or "settling" in sex partners, men may better gain access to the sexual gratification they are seeking while not imposing major costs on themselves. In contrast, because women pay heavier costs for making bad mating choices (Jonason et al., 2015), they may avoid this compensatory mating tactic. Beyond sex differences in the use of this tactic, we found that those who felt they were low in mate value and anxiously attached were likely to lower their standards for short- and long-term mates but these traits did not predict the adoption of the other tactics. This may mean that those who are characterized by these traits may have an immediate need to pair-up leading them to incur the costs associated with lowered standards.

17

We also examined the option of abstaining (Apostolou, 2017) whereby individuals may temporarily refrain from dating presumably in hopes of finding a better partner later. On its surface, this tactic seems like a bad choice because it could mean reproductive oblivion. However, if the abstention is temporary, it might be a useful tactic for such a long-lived species as humans. This tactic allows one to maintain relatively high standards unlike lowering one's standards, but may come with the costs of missed opportunities, loneliness, and reproductive exclusion. Such people may not fear being single (Spielmann et al., 2013). Both sexes appeared to favor this option for long-term mates, but not as much for short-term mates. If one's goals are short-term in nature, abstention undercuts the very objectives one is pursuing. That is, abstention as a compensatory mating tactic is a long-term solution; biding one's time now in hopes of finding a better option later (Apostolou, 2017, 2019). We found limited evidence for a role of mate value or attachment in describing individual differences in adopting this tactic. The little evidence we found suggests those with high in mate value as a long-term partner were less likely to abstain for a short-term mate whereas those with an avoidant attachment pattern were more likely to abstain for short-term mates. The former may reflect a belief in those with high mate value have no need to abstain because of their value whereas the latter may reflect an aversion of sexual intimacy for avoidant individuals.

And last, we examined the option of increasing the size of one's mating pool by traveling farther (Jonason, Nolland, & Tyler, 2017). This tactic allows people to offset the costs of the other two whilst incurring travel costs and the potential difficulty of engaging in a relationship at a distance (Carpenter & Knox, 1996; Feeney, 1999; Sahlstein, 2004). Both sexes invest heavily in long-term partnerships (Li et al., 2002) making having a high-quality partner especially important. By traveling farther to find new mates, both sexes avoid the costs associated with lowering one's standards while still finding a partner which would be delayed by abstention. In addition, we found that those with low mate value and an anxious

attachment pattern were willing to choose this compensatory tactic in the short- but not the long-term. This may reflect a willingness to do anything to satisfy their sexual needs in these people, anything including traveling farther, incurring those costs in the pursuit of sex.

As noted above, there is a substantial (Bech-Sørensen & Pollet, 2016; Chang et al., 2011) and contentious (Li & Metzler, 2015; Zetner & Eagly, 2015) literature surrounding questions regarding mating psychology. We have attempted to avoid that debate by not concerning ourselves with questions of what men and women want in their mates and why. Instead, we attempted to examine a more pragmatic issue, one that has direct relevance for people's relationship and sexual satisfaction. Nevertheless, our results are more consistent with evolutionary than sociocultural models of mating psychology. Sociocultural models have no *a priori* reasons (we know of) to predict the patterns we found. Instead, the asymmetries of reproductive costs afforded by modern evolutionary models of mating psychology allow for a more robust account of what we found.

Our results may have some applied implications. Many people struggle to satisfy their sexual and/or relationship agendas. They remain unpaired when they would rather be in a relationship or they remain celibate when they would rather engage in sex (e.g., "incels"). Our study directly informs people and counselors as to the pros and cons of three different solutions to mate shortages. While people may have already engaged in such tactics or even advised clients, friends, and family to do so, they may have not fully considered the ramifications of these tactics nor how those tactics may align or misalign with evolved biases based on parental investment theory (Trivers, 1972). We hope our results help others better consider their options when facing mate shortages.

# **Limitations and Conclusions**

Despite the novelty and methodological heterogeneity of our studies, several limitations are worth mentioning. First, although our samples were older in age than typical

college-student samples, our is still W.E.I.R.D. (i.e., Western, educated, industrialized, rich, and democratic; Henrich, Heine, & Norenzayan, 2010) and online. Although we have no particularly strong reason to distrust online samples (Buhrmester, Kwang, & Gosling, 2011), cross-cultural work might be warranted as Americans have ready access to cars and other public transportation than in developing nations that might attenuate the adoption of mating tactics like traveling and higher population density may permit abstention because the probability of meeting another partner is higher than in tribal societies (Ember, 1978).

Second, we may have constrained ourselves by focusing on only two relationship contexts and the specific individual differences we included. For instance, although comparing short-term and long-term relationships is useful for highlighting sex-based differences in mate preferences (Buss & Schmitt, 1993), a comprehensive account of what people want in their romantic/sexual partners must include a wider range of relationships (e.g., same-sex and polyamorous; see Jonason & Balzarini, 2016; Jonason, Valentine, & Li, 2012). Similarly, while we examined two individual differences that may influence decisions in relationships, others of relevance might include sociosexuality (Simpson & Gangestad, 1991), income, and approach/avoidance tendencies.

Third, we have examined hypothetical mating strategies as opposed to actual adjustments people might make in response to real mating shortages. This means that our results not only lack ecological validity but also participant's decisions could be described as "cold reasoning". Prior work has cast doubt on the trustworthiness of hypothetical judgments in the mating domain (Eastwick & Finkel, 2008), but subsequent and potentially superior work reveals that hypothetical choices resemble "real" choices (Li et al., 2013). Nevertheless, experimental manipulations of mate shortages are warranted. For instance, making people believe they are unlikely to find a mate in the near future and then assessing tactical choices might prove fruitful.

Fourth, we only explored the role of three tactics in solving the adaptive problem of mate shortages. There may be more tactics that people have developed over the millions of years of human evolution and ones that people may have inherited through phylogenetic inertia since the evolution of sex some 500 million years ago. For example, mate poaching might be a compensatory mating strategy where men and women can try to compensate for their inability to find a committed partner by merely having sex with people. That is, they lower their standards for commitment in exchange or sex. Indeed, such a mating strategy may come with relationship dissatisfaction as a cost (Foster et al., 2013). Nevertheless, our goal here was not to document every possible mating tactic; future research might explore this more fully. However, it is possible that there are social desirability effects (that we cannot control for) in people's willingness to choose a given tactic. We hope to have minimized this through the use of online methods.

Fifth, one might contend that the definitions we used to prompt participants with short-term and long-term contexts were problematic. We hoped to standardize responses and get participants thinking about specific relationships to improve the validity of the hypothetical task they were engaging in. Nevertheless, the use of specific relationship terms may have created some noise in our model, thereby attenuating our results.

Sixth, the omission of online dating as a potential tactic might be problematic. While there is a growing body of research on the use of dating applications and websites, we would contend that the mating tactics laid out in this study are larger in scope than adopting online dating. Online dating is a way to find mates, but the tactics people employ within this mating niche are likely characterized by the tactics we explored here. We expect that abstention (i.e., getting off the application), lowering one's standards (i.e., swiping more liberally), and searching farther (i.e., expanding one's search radius) play out in a similar fashion on and offline as humans carry their evolved mating psychology with them across locations and

21

contexts. Despite these limitations, we have detailed new information about mating tactics as they are geared towards solving the adaptive problem of mate shortages.

While there has been research on responses to perceived scarcity of mates or imbalances in the ratio of men to women in a population (Arncocky, Woodruff, & Schmitt, 2016; Arnocky et al., 2014; Taylor, 2013; Watkins et al., 2012) less research has examined the tactical responses people might adopt when suffering mate shortages. In this study, we examined what people do—in a tactical sense—when they cannot find a romantic/sexual partner. We explored three options, examined sex differences, and tested the role of individual differences in attachment and mate value. The results align with evolutionary models of mating consistent with parental investment theory (Trivers, 1972) because the choice of different compensatory tactics overall and in men and women appears sensitive to the costs/benefits associated with contextual differences in relationship context in men and women (Buss & Schmitt, 1993). While there may be many other potential tactics one adopts, we feel these three represent different decision-making processes and, therefore, represent a good cross-section of the responses people might adopt when facing mate shortages. We encourage more work that can inform on the processes that people engage in when adjusting their mating tactics in response to feedback in their dating lives.

#### References

- Ainsworth, M.D.S., Blehar, M.C., Waters, E., & Wall, S. (1978). *Patterns of attachment:*Assessed in the strange situation and at home. Hillsdale, NJ: Erlbaum.
- Apostolou, M. (2019). Why men stay single? Evidence from Reddit. *Evolutionary*Psychological Science, 5), 87-97.
- Apostolou, M. (2017). Why people stay single: An evolutionary perspective. *Personality and Individual Differences*, 111, 263–271.
- Arnocky, S., Ribout, A., Mirza, R., & Knack, J. M. (2014). Perceived mate availability influences intrasexual competition, jealousy and mate guarding behavior. *Journal of Evolutionary Psychology*, 12, 45–64.
- Arnocky, S., Woodruff, N.W., & Schmitt, D.P. (2016). Men's sociosexuality is sensitive to changes in mate-availability. *Personal Relationships*, 23, 172–181.
- Batholomew, K., & Horowitz, L.M. (1991). Attachment styles among young adults: A test of a four-category model. *Journal of Personality and Social Psychology*, 61, 226-244.
- Brennan, K.A., & Shaver, P.R. (1995). Dimensions of adult attachment, affect regulation, and romantic relationship functioning. *Personality and Social Psychological Bulletin, 21*, 267-283.
- Bech-Sørensen, J., & Pollet, T.V. (2016). Sex differences in mate preferences: A replication study, 20 years later. *Evolutionary Psychological Sciences*, 2, 171–176.
- Bossard, J.H.S. (1932). Residential propinquity as a factor in marriage selection. *American Journal of Sociology*, 38, 219-224.
- Buhrmester, M., Kwang, T., & Gosling, S.D. (2011). Amazon's Mechanical Turk: A new source of inexpensive, yet high-quality, data? *Perspectives on Psychological Science*, 6, 3–5.

- Buss, D.M., & Schmitt, D.P. (1993). Sexual strategies theory: An evolutionary perspective on human mating. *Psychological Review*, *100*, 204-232.
- Buss. D.M., & Shackelford, T.K. (2008). Attractive women want it all: Good genes, investment, parenting indicators, and commitment. *Evolutionary Psychology*, 6, 134-146.
- Carpenter, D., & Knox, D. (1996). Relationship maintenance of college students separated during courtship. *College Student Journal*, 28, 86-88.
- Chang, L., Wang, Y., Shackelford, T.K., & Buss, D.M. (2011). Chinese mate preferences:

  Cultural evolution and continuity across a quarter of a century. *Personality and Individual Differences*, 50, 678-683.
- Clarke, P.M.R., Henzi, S.P., Barrett, L., & Rendall, D. (2008). On the road again:

  Competitive effects and condition-dependent dispersal in male baboons. *Animal Behaviour*, 76, 55-63.
- Duguay, S. (2016). Dressing up Tinderella: Interrogating authenticity claims on the mobile dating app Tinder. *Information, Communication, & Society*, 20, 1-17.
- Eagly, A.H. (1987). Sex differences in social behavior: A social role interpretation. Hillsdale, NJ: Erlbaum.
- Eagly, A.H., & Wood, W. (1999). The origins of sex differences in human behavior: Evolved dispositions versus social roles. *American Psychologist*, *54*, 408-423.
- Eastwick, P.W., & Finkel, E.J. (2008). Sex differences in mate preferences revisited: Do people know what they initially desire in a romantic partner? *Journal of Personality and Social Psychology*, 94, 245–264.
- Ember, C.R. (1978). Myths about hunter-gatherers. Ethnology, 17, 439-448.
- Feeney, J. (1999). Issues of closeness and distance in dating relationships: Effects of sex and attachment style. *Journal of Social and Personal Relationships*, 16, 571-590.

- Feingold, A. (1992). Gender differences in mate selection preferences: A test of the parental investment model. *Psychological Bulletin*, *112*, 125-139.
- Fletcher, G.J.O., Tither, J.M., O'Loughlin, C., Friesen, M., & Overall, N. (2004). Warm and homely or cold and beautiful?: Sex differences in trading off traits in mate selection.

  \*Personality and Social Psychology Bulletin, 30, 659-672.
- Fisher, M., Cox, A., Bennett, S., & Gavric, D. (2008). Components of self-perceived mate value. *Journal of Social, Evolutionary, and Cultural Psychology*, 2, 156-168.
- Foster, J.D., Jonason, P.K., Shrira, I., Campbell, W.K., Shiverdecker, L.K., & Varner, S.C. (2014). What do you get when you make someone else's partner your own? An analysis of relationships formed via mate poaching. *Journal of Research in Personality*, 52, 78-90.
- Gutierres, S.E., Kenrick, D.T., & Partch, J.J. (1999). Beauty, dominance, and the mating game: Contrast effects in self-assessment reflect gender differences in mate selection.

  \*Personality and Social Psychology Bulletin, 25, 1126-1134.
- Harrison, A.A., & Saeed, L. (1977). Let's make a deal: An analysis of revelations and stipulations in lonely hearts advertisements. *Journal of Personality and Social Psychology*, *35*, 257-264.
- Hazan, C., & Shaver, P.R. (1987). Romantic love conceptualized as an attachment process. *Journal of Personality and Social Psychology*, 52, 511-524.
- Henrich, J., Heine, S J., & Norenzayan, A. (2010). The weirdest people in the world? Behavioral and Brain Sciences, 33, 61-83.
- Hoffman, J.I., Forcada, J., Trathan, P.N., & Amos, W. (2007). Female fur seals show active choice for males that are heterozygous and unrelated. *Nature*, 445, 912-914.

- Howard, J.A., Blumstein, P., & Schwartz, P. (1987). Social or evolutionary theories?: Some observations on preferences in human mate selection. *Journal of Personality and Social Psychology*, *53*, 194-200.
- Fraley, R.C., Waller, N.G., & Brennan, K.A. (2000). An item-response theory analysis of self-report measures of adult attachment. *Journal of Personality and Social Psychology*, 78, 350-365.
- Jonason, P.K. (2013). Four functions for four relationships: Consensus definitions in university students. *Archives of Sexual Behavior*, 42, 1407-1414.
- Jonason, P.K., & Balzarini, R. (2016). Unweaving the rainbow of human sexuality. In K. Aumer (Ed.) *The psychology of love and hate in intimate relationships* (pp. 13-28). New York, NY: Springer.
- Jonason, P.K., Garcia, J., Webster, G.D., Li, N.P., & Fisher, H. (2015). Relationship dealbreakers: What individuals do not want in a mate. *Personality and Social Psychological Bulletin*, 41, 1697-1711.
- Jonason, P.K., Marsh, K., Dib, O., Plush, D., Doszpot, M., Fung. E., Crimmins, K., Drapski,
  M., & Di Pietro, K. (2019). Is smart sexy?: Examining the role of relative intelligence
  in mate preferences. *Personality and Individual Differences*, 139, 53-59.
- Jonason, P.K., Nolland, M., & Tyler, M.D. (2017). Incorporating geographic distance into mate preference: Necessities and luxuries of mate preferences. 2.0. *Personal Relationships*, 24, 585-597.
- Jonason, P.K., Valentine, K.A., & Li, N.P. (2012). Human mating. In V. S. Ramachandran (Ed.), *Encyclopedia of Human Behavior* (2nd ed., Vol. 2, pp. 371-377). Oxford, England: Academic Press.
- Kenrick, D.T., Groth, G.E., Trost, M.R., & Sadalla, E.K. (1993). Integrating evolutionary and social exchange perspectives on relationships: Effects of gender, self-appraisal, and

- involvement level on mate selection criteria. *Journal of Personality and Social Psychology*, 64, 951-969.
- Kirsner, B.R., Figueredo, A.J., & Jacobs, W.J. (2003). Self, friends, and lovers: structural relations among Beck Depression Inventory scores and perceived mate-values.

  \*\*Journal of Affective Disorders, 75, 131–138.\*\*
- Li, N.P., Bailey, J.M., Kenrick, D.T., & Linsenmeier, J.A.W. (2002). The necessities and luxuries of mate preferences: Testing the tradeoffs. *Journal of Personality and Social Psychology*, 82, 947-955.
- Li, N.P., & Kenrick, D.T. (2006). Sex similarities and differences in preferences for short-term mates: What, whether, and why. *Journal of Personality and Social Psychology*, 90, 468-489.
- Li, N.P., & Meltzer, A.L. (2015). The validity of sex-differentiated mate preferences:

  Reconciling the seemingly conflicting evidence. *Evolutionary Behavioral Sciences*, 9, 89-106.
- Li, N.P., Valentine, K.A., & Patel, L. (2011). Mate preferences in the U.S. and Singapore: A cross-cultural test of the mate preference priority model. *Personality and Individual Differences*, 50, 291-294.
- Li, N.P., Yong, J.C., Tov, W., Sng, O., Fletcher, G.J.O., Valentine, K.A., Jiang, Y.F., & Balliet, D. (2013). Mate preferences do predict attraction and choices in the early stages of mate selection. *Journal of Personality and Social Psychology*, 105, 757-776.
- Mikulincer, M., & Shaver, P.R. (2003). The attachment behavioral activation system in adulthood: Activation, psychodynamics, and interpersonal processes. In M.P. Zanna (Ed.), *Advances in experimental social psychology* (Vol. 35). San Diego, CA: Academic Press.

- Moss, J.H., & Maner, J.K., (2015). Biased sex ratios influence fundamental aspects of human mating. *Personality and Social Psychologoical Bulletin*, 42, 72-80.
- Reeve, S.D., Kelly, K.M., & Welling, L.L. (2016). Transitory environmental threat alters sexually dimorphic mate preferences and sexual strategy. *Evolutionary Psychological Science*, 2, 101-113.
- Regan, P.C. (1998). What if you can't get what you want?: Willingness to compromise ideal mate selection standards as a function of sex, mate value, and relationship context.

  Personality and Social Psychology Bulletin, 24, 1294-1303.
- Sahlstein, E.M. (2004). Relating at a distance: Negotiating being together and being apart in long-distance relationships. *Journal of Social and Personal Relationships*, 21, 679-710.
- Schachner, D.A., & Shaver, P.R. (2004). Attachment dimensions and sexual motives.

  \*Personal Relationships, 11, 179-195.
- Schacht, R., & Borgerhoff Mulder, M. (2015). Sex ratio effects on reproductive strategies in humans. *Royal Society Open Science*, 2. https://doi.org/10.1098/rsos.140402
- Schmitt, D.P., & Jonason, P.K. (2015). Attachment and sexual permissiveness: Exploring differential associations across genders, cultures, and facets of short-term mating. *Journal of Cross-Cultural Psychology*, 46, 119-133.
- Schönbrodt, F.D., & Perugini, M. (2013). At what sample size do correlations stabilize? *Journal of Research in Personality*, 47, 609-612.
- Shaver, P.R., & Brennan, K.A. (1992). Attachment styles and the "big five" personality traits:

  Their connections with each other and with romantic relationship outcomes.

  Personality and Social Psychology Bulletin, 18, 536-545.
- Simpson, J.A. (1990). Influence of attachment styles on romantic relationships. *Journal of Personality and Social Psychology*, 59, 971-980.

- Simpson, J.A., & Gangestad, S.W. (1991). Individual differences in sociosexuality: Evidence for convergent and discriminant validity. *Journal of Personality and Social Psychology*, 60, 870-883.
- Spielmann, S.S., MacDonald, G., Maxwell, J.A., Joel, S., Peragine, D., Muise, A., & Impett,
  E.A. (2013). Settling for less out of fear of being single. *Journal of Personality and Social Psychology*, 105, 1049-1073.
- Symons, D. (1979). *The evolution of human sexuality*. New York, NY: Oxford University Press.
- Taylor, L.D. (2013). Male partner selectivity, romantic confidence, and media depictions of partner scarcity. *Evolutionary Psychology*, 11, 36-49.
- Trivers, R.L. (1972). Parental investment and sexual selection. In B. Campbell (Ed.), *Sexual selection and the descent of man* (pp. 136-179). Chicago, IL: Aldine.
- Watkins, C.D., Jones, B.C., Little, A.C., Debruine, L., & Feinberg, D.R. (2012). Cues to the sex ratio of the local population influence women's preferences for facial symmetry. *Animal Behaviour*, 83, 545–533.
- Wilkins, J.S., & Ebach, M.C. (2014). Homology and analogy. In J.S. Wilkin & M.C. Ebach (Eds.), *The nature of classification* (pp. 85-108). London, England: Palgrave Macmillan.
- Zentner, M., & Eagly, A.H. (2015). A sociocultural framework for understanding partner prefences of women and men: Integration of concepts and evidence. *European Review of Social Psychology*, 26, 328-373.

# **TABLES**

Table 1. Overall choice in each tactic and sex differences in those choices (Study 1).

		Count (%)	$\chi^2$	Φ	
<b>Long-Term Mate</b>	Overall	Men	Women	_	
Abstaining	73 (24)	41 (25)	32 (24)		
Lowering standards	43 (14)	26 (16)	17 (13)	0.90	.04
Travelling farther	183 (61)	98 (59)	85 (63)		
<b>Short-Term Mate</b>					
Abstaining	131 (44)	47 (29)	84 (63)		
Lowering standards	68 (23)	55 (33)	13 (10)	52.17**	.34
Travelling farther	100 (34)	63 (38)	37 (28)		

<sup>\*</sup> *p* < .05, \*\* *p* < .01

Table 2. Descriptive statistics for responses to mating shortages (Study 2).

	Traveling	Lowering	
	Farther	Standards	Abstaining
Overall STM	2.49 (1.22)	2.26 (1.17)	2.84 (1.35)
Men	2.54 (1.22)	2.58 (1.14)	2.65 (1.39)
Women	2.46 (1.22)	2.04 (1.14)	2.96 (1.32)
Overall LTM	2.73 (1.18)	1.96 (1.05)	2.95 (1.33)
Men	2.64 (1.15)	2.12 (1.05)	2.83 (1.36)
Women	2.80 (1.20)	1.84 (1.03)	3.02 (1.30)

*Note*. STM = short-term mating; LTM = long-term mating

Table 3. Correlations between mate value and attachment and likelihood of choosing each mating tactic in response to mating shortages (Study 2).

-	Traveling Farther		Lowering Standards			Abstaining			
	STM	LTM	z	STM	LTM	z	STM	LTM	z
STM Attractiveness	.05	07	2.18*	01	.06	-1.33	.02	06	1.58
LTM Attractiveness	.07	.14*	-1.27	13*	08	-0.95	03	14*	2.19*
General Undesirability	.15**	.06	1.64	.24**	.31**	-1.40	.05	.09	-0.79
Attachment Anxiety	.13*	.04	1.64	.21**	.29**	-1.58	01	.03	-0.79
Attachment Avoidance	11	07	-0.72	01	01	0.00	.14*	.09	0.99

*Note.* z Steiger's z to compare dependent correlations; STM = short-term mating; LTM = long-term mating \* p < .05, \*\* p < .01